Author's response to reviews

Title: Prevalence of diabetic retinopathy in Tehran Province: A population-based study

Authors:

Mohammad Ali Javadi (labbafi@hotmail.com)
Marzieh Katibeh (mdkatibeh@yahoo.com)
Nasrin Rafati (labbafi@hotmail.com)
Mohammad Hossein Dehghan (Mhdehghan5@hotmail.com)
Farid Zayeri (labbafi@hotmail.com)
Mehdi Yaseri (m.yaseri@yahoo.com)
Mojtaba Sehat (om_sehat@yahoo.com)
Hamid Ahmadieh (hahmadieh@hotmail.com)

Version: 8 Date: 10 October 2009

Author's response to reviews: see over
Dear editorial team

Thank you for checking the manuscript’s format and accepting it. I checked the last version of manuscript again and changed it a little as follow:

1. In the abstract section the response rate should change from 84.1 to 84.2 (actually it was 84.18)

2. One parenthesis changed to a bracket.

3. In the footnote part of table 1 "95% confidence intervals" change to "confidence intervals 95%"

      I would confirm that manuscript is ready to publish now.

Sincerely

Marzieh Katibeh, MD, MPH
Corresponding Author
Telephone: (98)21-22766106
Fax: (98) 21-22587317
Email: mdkatibeh@yahoo.com